Author’s response to reviews

Title: Role of biomechanics in the understanding of normal, injured, and healing ligaments and tendons

Authors:

Ho-Joong Jung (sunu@cau.ac.kr)
Matthew B. Fisher (mbf7@pitt.edu)
Savio L-Y. Woo (ddecenzo@pitt.edu)

Version: 4 Date: 12 May 2009

Author’s response to reviews:

This is the final proof. Gramatical errors have been corrected in the text, and Figure 5 has been slightly modified to improve its appearance.